Author’s response to reviews

Title: IL-10+ NK and TGF-β+ NK cells play negative regulatory roles in HIV infection

Authors:

Yongjun Jiang (jiangjun55555@163.com)
Mei Yang (1285175459@qq.com)
Xiaojuan Sun (364010041@qq.com)
Xi Chen (chenxi9204@yeah.net)
Meichen Ma (maomaochong432@163.com)
XiaoWan Yin (yinxiaowan1013@163.com)
Shi Qian (poem.z@163.com)
Zining Zhang (zi_ning101@hotmail.com)
Yajing Fu (fufu80s@sina.com)
Jing Liu (lj_sycmu@163.com)
Xiaoxu Han (xxhan@yeah.net)
Junjie Xu (xjjbeijing@gmail.com)
Hong Shang (hongshang100@hotmail.com)

Version: 3  Date: 31 Jan 2018

Author’s response to reviews:

Question 1) Please clarify the contribution of JX who is listed as an author but not mentioned in the authors’ contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

Response:

We apologized for the typo mistake, we have revised “YJ” into “XJ” in the line 320 on page 15.
Question 2) Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response:
We asked all the authors to read this revised manuscript and get the approval, and we have added this following statement “all authors have read and approved the manuscript” to the Authors' contributions section in the line 321-322 on page 15.

Question 3) Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file. Please ensure you include all the figures/titles cited in the manuscript as, currently, Figures 1-8 are missing.

Response:
We have re-uploaded all the figures cited in the manuscript.

Question 4) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response:
We have re-uploaded tables/figures/manuscript as a single, final, clean version.